Impact of real-time metabolomics in liver transplantation: Graft evaluation and donor-recipient matching

Graphical abstract



Highlights

- Real-time metabolomics by HR-MAS-NMR accurately predicts early allograft dysfunction.
- Lactate and phosphocholine content are highly predictive biomarkers of early allograft dysfunction.
- Livers from patients with sarcopenia and cirrhosis have low lactate and phosphocholine content.
- Metabolic adaptation in these patients may lead to poor short-term outcomes.

Authors

Francois Faitot, Camille Besch, Stephanie Battini, ..., Bernard Ellero, Philippe Bachellier, Izzie-Jacques Namer

Correspondence

izzie.jacques.namer@chru-strasbourg.fr (I.-J. Namer)

Lay summary

Real-time metabolomic profiles of human grafts during back-table can accurately predict graft dysfunction. High lactate and phosphocholine content are highly predictive of graft dysfunction whereas low lactate and phosphocholine content characterize patients with sarcopenia. In these patients, the cost of metabolic adaptation may explain the poor outcomes.



Impact of real-time metabolomics in liver transplantation: Graft evaluation and donor-recipient matching

Francois Faitot^{1,2}, Camille Besch¹, Stephanie Battini², Elisa Ruhland², Mihaela Onea³, Pietro Addeo¹, Marie-Lorraine Woehl-Jaeglé¹, Bernard Ellero¹, Philippe Bachellier¹, Izzie-Jacques Namer^{2,4,*}

¹Hepatobiliopancreatic Surgery and Transplantation Department, Hopital de Hautepierre, CHU de Strasbourg, France; ²Laboratoire ICube, UMR7357, University of Strasbourg, France; ³Pathology Department, Hopital de Hautepierre, CHU de Strasbourg, France; ⁴Nuclear Medicine Department, Hôpital de Hautepierre, CHU de Strasbourg, France

Background & Aims: There is an emerging need to assess the metabolic state of liver allografts especially in the novel setting of machine perfusion preservation and donor in cardiac death (DCD) grafts. High-resolution magic-angle-spinning nuclear magnetic resonance (HR-MAS-NMR) could be a useful tool in this setting as it can extemporaneously provide untargeted metabolic profiling. The purpose of this study was to evaluate the potential value of HR-MAS-NMR metabolomic analysis of back-table biopsies for the prediction of early allograft dysfunction (EAD) and donor-recipient matching.

Method: The metabolic profiles of back-table biopsies obtained by HR-MAS-NMR, were compared according to the presence of EAD using partial least squares discriminant analysis. Network analysis was used to identify metabolites which changed significantly. The profiles were compared to native livers to identify metabolites for donor-recipient matching.

Results: The metabolic profiles were significantly different in grafts that caused EAD compared to those that did not. The constructed model can be used to predict the graft outcome with excellent accuracy. The metabolites showing the most significant differences were lactate level >8.3 mmol/g and phosphocholine content >0.646 mmol/g, which were significantly associated with graft dysfunction with an excellent accuracy (AUROC_{lactates} = 0.906; AUROC_{phosphocholine} = 0.816). Native livers from patients with sarcopenia had low lactate and glycerophosphocholine content. In patients with sarcopenia, the risk of EAD was significantly higher when transplanting a graft with a high-risk graft metabolic score.

Conclusion: This study underlines the cost of metabolic adaptation, identifying lactate and choline-derived metabolites as predictors of poor graft function in both native livers and liver grafts. HR-MAS-NMR seems a valid technique to evaluate graft quality and the consequences of cold ischemia on the graft. It could be used to assess the efficiency of graft resuscitation on machine perfusion in future studies.

E-mail address: izzie.jacques.namer@chru-strasbourg.fr (I.-J. Namer).



Lay summary: Real-time metabolomic profiles of human grafts during back-table can accurately predict graft dysfunction. High lactate and phosphocholine content are highly predictive of graft dysfunction whereas low lactate and phosphocholine content characterize patients with sarcopenia. In these patients, the cost of metabolic adaptation may explain the poor outcomes. © 2017 European Association for the Study of the Liver. Published by Elsevier B.V. All rights reserved.

Introduction

Liver transplantation is a life-saving procedure for patients with end-stage liver disease and a potentially curative treatment for hepatocellular carcinoma. The major limitation for liver transplantation is the current organ shortage caused by an increasing discrepancy between indications and a stable donor pool. In an attempt to answer to the issue of the increasing number of patients on waiting lists, many teams have extended the criteria for acceptance of liver grafts. Although there is hardly a wide consensus on its definition, extended criteria donors (ECD) represent a growing proportion of the donors. Many donor factors have been reported as influencing the outcome of liver transplantation mainly age,¹ steatosis^{2,3} and cold ischemia time.⁴ While ECD, particularly DCD, has increased the pool of donors, it may be associated with higher graft loss⁵ or increased risk of vascular and biliary complications.^{6,7} Indeed the tolerance of allograft to cold ischemia-reperfusion injury is altered in ECD allografts.^{8,9}

Donor-recipient matching is of utmost importance in this setting. Model for end-stage liver disease (MELD) score and life-support therapies have been identified as significant recipient factors that alter results, specifically when using extended criteria donor. The balance of risk (BAR) score is an example of application of such donor-recipient matching. Additionally, the addition of graft steatosis in the score further enhances the accuracy of the BAR score.¹⁰

Among significant factors impacting early outcomes after liver transplantation, sarcopenia and portal hypertension have been increasingly studied in the last years. Both are probably linked, as ascites is often associated with malnutrition, and portal hypertension may be a surrogate of long-term evolving cirrhosis. Sarcopenia may be a marker of significant metabolic shift and there are currently no clear data on differentially

Keywords: Liver transplantation; Metabolomics; Lactate; Early allograft dysfunction. Received 15 June 2017; received in revised form 9 November 2017; accepted 15 November 2017; available online 2 December 2017

^{*} Corresponding author. Address: Nuclear Medicine Department, CHU Strasbourg; ICube laboratory, UMR7357, University of Strasbourg, France. Tel.: +33 88128650; fax: +33 8127286.

Research Article

expressed metabolites in the liver of patients with cirrhosis and sarcopenia.

There is currently a lack of effective tools and biomarkers to evaluate the liver grafts before implantation. Liver biopsies with fibrosis and steatosis assessment may be informative,¹¹ but do not take into consideration metabolic insults caused by static cold storage. Interpretation of frozen section biopsies can be confusing and can mislead clinicians who have to decide whether to use marginal grafts.¹² In order to further extend and enhance the quality of the grafts, machine perfusion is currently under intense evaluation for liver grafts.^{13–15} The benefit of dynamic cold storage has been demonstrated in kidney transplantation. The data available tend to show efficient "resuscitation" of ECD grafts especially steatotic grafts.^{17,18}

Metabolomics is an emerging area in the omics field consisting of the simultaneous evaluation of cellular metabolic products on liquid or solid phase. Nuclear magnetic resonance (NMR) spectroscopy metabolomics has already been applied in the field of liver transplantation,¹⁹ but only mass spectroscopy has been shown to be predictive of allograft dysfunction.²⁰ Whereas many metabolomic methods are not relevant for clinical practice because they need complex sample handling and long treatment time, 1H high-resolution magic-angle-spinning NMR (1H HR-MAS NMR) spectroscopy is an attractive solution. Metabolomic profile of solid fresh frozen biopsy can be obtained in a short period of time without destruction of the sample, thus enabling further classical histopathological evaluation.

The goal of this study was to evaluate the potential value of 1H HR-MAS NMR metabolomics in a clinical setting and eventually identify biomarkers in the graft and in the native liver to predict early outcomes after liver transplantation. The main hypothesis was that the liver allograft exerts significant metabolic derangements that depend on each allograft capacity to cope with cold ischemia. At the early stages of reperfusion, the allograft is exposed to significant changes in metabolism, depending on the metabolic state of the recipient. Metabolic donor-recipient matching may be a novel way of looking at early graft function.

Material and methods

This study included patients transplanted between December 2014 and December 2016 who responded to the following criteria: first liver transplantation, absence of early vascular complications, available snap-frozen biopsy within less than 5 min after realization of the biopsy, available histopathological analysis of the specimen, available biological and clinical data and informed consent from the patient. The Ethics Committee affiliated to Strasbourg University (Comité de Protection des Personnes "Est IV") approved the study (Registration number: 09/39 a).

Design of the study

The value of metabolomic profiling using HR-MAS-NMR was evaluated on a set of 42 liver grafts at the time of *ex vivo* preparation and 36 native livers. To check whether the studied population was representative, donor and recipient characteristics of the population transplanted during the same period of time but not included (n = 92) were compared to the patients included.

In a first step, a multivariate non-targeted analysis was used to compare metabolic profiles according to the occurrence of early allograft dysfunction (EAD). Network analysis was conducted to identify differentially expressed metabolites between EAD and non EAD patients as well as between recipients with or without sarcopenia. Optimal cut-off values for the most accurate biomarkers were then defined. A liver graft metabolic score (GMS) using these parameters was calculated. In a second step, this score was evaluated for prediction of EAD and one-year graft loss. Predictive factors for high metabolite levels were searched to define ECD using available clinical data from the donor chart.

Surgical technique and sample procurement

Back-table biopsies are routinely performed at the beginning of the back-table in our center, without selection criteria according to the donor characteristics, but only biopsies that were rapidly snap-frozen in nitrogen were analyzed in order to avoid metabolic changes due to ischemia. A biopsy from the native liver was also procured just after portal clamping.

All donors were brain dead donors procured according to a classical technique.²¹ Portal cannulation with washing was always performed using Ringer-lactate before clamping and dual arterial and portal washing was achieved with various conservation liquids.

Histopathological analysis was performed on the sample analyzed in spectroscopy to take into consideration zonal variations of necrosis, fibrosis and steatosis.

ECD were defined according to the European Association for the Study of the Liver (EASL) guidelines²² as the donor-risk index is not adapted to the liver transplantation organization in France.

Liver recipients were listed on the French national list after complete pre-transplantation evaluation. Indication, presence of hepatocellular carcinoma and lab-MELD at listing and at transplantation are reported.

Endpoint definition

EAD was defined according to the Olthoff's criteria,²³ namely bilirubin \geq 170 µmol/L at day seven, INR \geq 1.6 at day seven and peak aminotransferases \geq 2,000 within the first seven postoperative days, in the absence of technical causes.

Sarcopenia was defined as low psoas surface as defined by Golse *et al.* according to the psoas surface.²⁴

Metabolomic study

Samples were prepared in a -20 °C environment. A total of 15 to 20 mg of tissue were punched-biopsied from the specimen. Deuterium oxide (8µl) with 0.75 weight percent 2,2,3,3-D4-3- (trimethylsilyl) propionic acid was added for chemical shift reference for NMR spectrometer.

HR-MAS analysis was achieved on a Brucker Avance III 500 spectrometer operating at a proton frequency of 500.13 MHz and equipped with a 4 mm triple resonance gradient HR-MAS probe. The analysis was conducted at -80 °C after placing the insert in a 4 mm ZrO₂ rotor.

A one-dimensional (1 D) proton spectrum using Carr-Purcell-Meiboom-Gill (CPMG) pulse sequence and 128 transients was acquired for each tissue sample. Free induction decay was multiplied by an exponential window function of 0.3 Hz prior to Fourier transformation and the result was corrected for phase and baseline distortion using TopSpin 3.2 (Brucker GmbH, Germany). The chemical shift was calibrated to the peak of the methyl proton of L-lactate at 1.33 ppm. In order to confirm resonance assignments, two-dimensional (2D) heteronuclear experiments were recorded immediately after the end of 1 D spectra acquisition for four representative samples. Spectra were referenced by setting lactate doublet chemical shift to 1.33 ppm in proton dimension and 22.70 ppm in carbon dimension. Metabolites were assigned using standard metabolite chemical shift tables available in the literature.

Metabolite quantification was performed using an external reference standard of lactate (3 μ mol), scanned under the same analytical conditions. Spectra were normalized according to sample weight. Peaks of interest were automatically defined by an in-house program using Matlab 7.0 (Mathwork, Natik, USA). Peak integration was then compared to the one obtained with the lactate reference and corrected according to the number of protons. Only well-defined peaks with no overlapping in the 1 D CPMG spectra were selected for quantification.

Statistical analysis

Continuous variables are expressed as mean ± standard deviation. Student's *t* test or Mann-Whitney U test were used to compare the means between groups as appropriate. Categorical variables are expressed as number and percentage. Chi-square test was used to compare the distribution of categorical variables between groups. Spearman's test was performed to determine correlation between variables. These statistical analyses were performed using Statview software (USA).

Receiving Operating Characteristics (ROC) curves were constructed to evaluate the value of identified metabolic biomarkers and to identify the best cut-off for these variables. SPSS software was used for this analysis.

Statistical analysis for metabolomics study

Principal component analysis (PCA) was performed to evaluate whether metabolic profile differentiated groups of patients. The two measurements of model quality were $R^2\gamma$ and Q^2 representing the accuracy of fit (*i.e.* data variation) and accuracy of prediction respectively. $Q^2 \ge 0.5$ was considered a good predictor.

Network analysis

Algorithm to determine expected metabolite level alterations using mutual information (ADEMA) was applied on metabolite quantification values. This method enables strong and valid comparison of very small samples (2 vs. 2 as demonstrated in the original article).²⁵ The network was constructed using Kyoto Encyclopedia of Genes and Genomes and Selway's work²⁶ using the following pathways: glucose/lactate; glucose/ascorbate/glu tathione/glutamate; glucose/alanine/valine/isoleucine; choline/ glycerophosphocholine/phosphocholine/total choline.

For further details regarding the materials used, please refer to the CTAT table.

Results

Population

There were 42 available biopsies for the analysis. As shown (Table 1) 69% of donors were ECD according to EASL. Median donor age was 56.5 (21–81). Regarding recipients, there were 33% (n = 14) with very high MELD (>35) and 15% were transplanted while in the ICU. The main indication was alcohol cirrhosis (62%) followed by HCV and metabolic cirrhosis.

The incidence of EAD was 17% in the whole population and 21% in the ECD. The 90-day mortality was 4%.

The mean acquisition time was 12 min and the total analysis including sample preparation and metabolite quantification was 30 min.

Metabolomic profile from ex vivo liver graft predicts EAD

Metabolomic profiles were significantly different between *ex vivo* biopsies of liver grafts presenting EAD *vs.* no EAD ($Q^2 = 0.573$, $R^2\gamma = 0.697$) (Fig. 1). Of note cold ischemia time >6 h was not associated with a significantly different metabolic profile.

Metabolomic study identifies potential biomarkers of EAD

The network analysis showed higher lactate, glutamate, glutamine, alanine, valine, isoleucine and choline derivatives concentration in EAD grafts. There was no change in glucose, ascorbate or GSH levels (Fig. 2).

Table	1.	Population	characteristics	and	comparison	to	non-selected
popula	atio	on during the	e same period of	f time			

	n = 42	n = 92	p value
Recipient's characteristics			
Age	53 ± 12	54 ± 11	0.646
Male gender	30 (71%)	63 (68%)	0.731
Indication for transplantation			0.166
Alcoholic	26 (62%)	46 (50%)	
HCV	5 (12%)	11 (12%)	
HBV	2 (5%)	0	
Metabolic	4 (10%)	8 (9%)	
Fulminant hepatitis	1 (2%)	6 (7%)	
Other	7 (17%)	22 (24%)	
Hepatocellular carcinoma	12 (29%)	26 (28%)	0.971
Lab-MELD at listing	26.8 ± 12	24.8 ± 11	0.371
Lab-MELD at LT	27.1 ± 14	22.2 ± 9	0.087
Bilirubin at LT (µmol/L)	192 ± 33	137 ± 21	0.114
Creatinine at LT (µmol/L)	83 ± 8	88 ± 7	0.634
INR at LT	2.92 ± 2	2.37 ± 1	0.064
Sarcopenia	14 (33%)	32 (35%)	0.87
ACLF	19 (45%)	27 (29%)	0.072
Donor's characteristics			
Age	57.7 ± 18	57.8 ± 17	0.972
BMI	26.4 ± 6	26.5 ± 5	0.891
Diabetes	7 (17%)	12 (13%)	0.594
Statin use	9 (21%)	25 (27%)	0.458
Metabolic syndrome	12 (29%)	16 (28%)	0.956
AST/ALT (IU)	$74 \pm 14/72 \pm 20$	$63 \pm 8/49 \pm 6$	0.507
Bilirubin (µmol/L)	14 ± 11	12 ± 6	0.472
GGT (IU)	59 ± 9	50 ± 5	0.354
Lactate at procurement (mmol/L)	1.8 ± 1.3	1.7 ± 1.3	0.819
Graft steatosis			
Overall degree of steatosis (%)	2% (0-65)	7% (0-25)	0.749
Macrovascular steatosis >30%	2 (6%)	0 (0%)	
Extended criteria donor*	29 (69%)	55 (58%)	0.304
Operative data			
Cold ischemia time	453 ± 91	462 ± 91	0.588
Reperfusion syndrome	20 (48%)	28 (30%)	0.212
Red blood cell transfusion (units)	7 ± 1	7 ± 1	0.973
Fresh frozen plasma (units)	9.5 ± 2	10 ± 1	0.68
Platelets (units)	1.3 ± 2	1.5 ± 2	0.713

ACLF, acute-on-chronic liver failure; ALT, alanine aminotransferase; AST, aspartate aminotransferase; GGT, gamma-glutamyl transferase; HBV, hepatitis B virus; HCV, hepatitis C virus; INR, international normalized ratio; LT, liver transplant; MELD, model for end-stage liver disease.

^{*}According to the European Association for the Study of the Liver definition.



Fig. 1. PCA analysis comparing metabolic profiles of back-table liver allograft biopsies in patients experiencing early allograft dysfunction (black dots; n = 7) or not (white dots; n = 35). This analysis shows a clear distinction of metabolic profiles between the two groups ($Q^2 > 0,5$). EAD, early allograft dysfunction; PCA, principal component analysis. PCA analysis with 1D and 3D representation.



Fig. 2. Box plot figures and bivariate analysis of lactate and phosphocholine content. (A) Box plot figure shows the significant difference in lactate and phosphocholine content according to the occurrence of EAD or not (*paired Student's *t* test). (B) Bivariate analysis of these variables according to the occurrence of EAD (blue dots) shows the individual values and the diagnostic value of phosphocholine and lactate. EAD, early allograft dysfunction.

Quantification showed that grafts presenting EAD had significantly higher lactate level (p < 0.0001), phosphocholine (p = 0.006) and taurine levels (p = 0.011).

Among the identified metabolites, only lactate and phosphocholine showed high accuracy for predicting EAD. ROC curve analysis showed an excellent predicting value of intragraft lactate level (AUROC = 0.906; 95% CI 0.813–0.999) (Fig. 3a). The optimal threshold was 8.3 mmol/g predicting EAD with a 100% sensitivity and 80% specificity. Similarly, AUROC for phosphocholine was 0.816 (95% CI 0.679–0.954) and a threshold of 0.65 mmol/g gave an 86% sensitivity and 80% specificity (Fig. 3b).

Predicting factors and prognostic value of MD-ECD

A metabolomic-defined extended criteria donor (MD-ECD) could be defined by the association of high lactate and high phosphocholine levels at the time of *ex vivo* preparation. None of the donor characteristics predicted MD-ECD, even

the degree of macroscopic steatosis. Neither cold ischemia time nor preservation solution were associated with MD-ECD.

MD-ECD were associated with a 63% risk of EAD. Most interestingly, MD-ECD recipients had a significantly higher risk of one-year graft loss and/or patient death (n = 3/8; 38%) than non MD-ECD (n = 3/34; 10%) (p = 0.037). In the MD-ECD, one graft was lost due to the only case of primary non-function (PNF), there were two deaths, one due to pulmonary sepsis leading to multiorgan failure at day 25, and one due to overwhelming biliary infection in the setting of ischemic cholangitis at 10 months post-transplantation. At reperfusion, MD-ECDs had higher levels of IL6 (1,309 ± 333 vs. 575 ± 108; p = 0.014).

Predicting factors of EAD

Patients presenting EAD had a significantly higher MELD ($37 \pm 7 vs. 25 \pm 13$; p = 0.014), with higher bilirubin and INR

JOURNAL OF HEPATOLOGY



Fig. 3. ROC curve analysis showing the sensitivity and specificity of intragraft lactate and phosphocholine content quantified by ¹H-HR-MAS-NMR. AUROC was of 0.906 for lactate content and 0.816 for phosphocholine content. AUROC, area under the ROC curve; ROC, receiver operating characteristic; Sen, sensitivity; Spe, specificity.

levels, less frequently HCC. Sarcopenia was associated with EAD (p = 0.019). In multivariate analysis, none of the donor characteristics predicted EAD. Cold ischemia time was not associated with EAD in this population (p = 0.57) (Table 2).

Multivariate analysis for EAD showed that graft lactate content was the only independent predictor of EAD (p = 0.046) (Table 2).

Recipient parameters affecting the metabolic profiles

Metabolomic profiles from patients with cirrhosis were compared according to MELD score, acute-on-chronic liver failure and presence of sarcopenia. Metabolomic profiles differed only according to presence or absence of sarcopenia ($Q^2 = 0.528$, $R^2\gamma = 0.783$).

Network analysis showed a lower level of lactate and glycerophosphocholine, but a significantly higher level of choline, neoglucogenic amino acids and ethanolamine. Using ROC curve analysis, only low glycerophosphocholine <0.556 mmol/g was an accurate biomarker (AUROC = 0.812) (Fig. 4).

Table 2.	Risk factors o	t early allograft	dysfunction in	uni- and i	nultivariate
analysis	6.				

	No EAD n = 35	EAD n = 7	p univariate	p multi
Recipient's characteristics				
Age	53 ± 12	53 ± 15	0.885	
Male gender	25 (71%)	5 (71%)	>0.999	
Indication for transplantation	20 (7 1/0)	0 (11,0)	0.000	
Alcoholic	23 (66%)	3 (43%)	0.256	
HCV	5 (14%)	0 (0%)	0.287	
Metabolic	3 (9%)	1 (14%)	0.638	
Hepatocellular carcinoma	12 (34%)	0 (0%)	0.067	
Lab-MELD at listing	24.7 ± 12	37.3 ± 7	0.013	
Lab-MELD at LT (uncapped)	24.3 ± 13	40.7 ± 11	0.003	0.116
Bilirubin at LT (umol/L)	162 ± 32	337 ± 103	0.041	
Creatinine at LT (umol/L)	83 ± 50	81 ± 37	0.905	
INR at LT	2.61 + 1	4.37 + 2	0.011	
Sarcopenia	9 (26%)	5 (71%)	0.019	0.158
Donor's characteristics	- ()	- ()		
Age	59 ± 17	53 ± 20	0.421	
BMI	26.7 + 6	24.8 + 6	0.468	
Diabetes	7 (20%)	0 (0%)	0 195	
Statin use	8 (23%)	1 (14%)	0.133	
Metabolic syndrome	11 (31%)	1 (14%)	0.359	
GGT (III)	55 + 9	82 + 32	0.333	
Lactate at harvesting (mmol/L)	178 ± 1	1 93 + 1	0.274	
Craft staatosis	1.70 ± 1	1.55 ± 1	0.700	
Overall steatosis >30%	0 (0%)	2 (28%)	0.002*	0 102
Macrovaccular staatosis (%)	57+2	2 (20%)	0.002	0.102
Extended criteria depor	J.7 ± 2	F (%)	0.137	
	25 (%)	0(%)	0.290	
Cold ischomia time (minutes)	452 ± 00	450 ± 104	0.026	
Ischemia time to bionsy	433 ± 90	4JU ± 104	0.920	
(minutes)	509 ± 74	512 ± 50	0.937	
Reperfusion syndrome	16 (46%)	4 (57%)	0.581	
Red blood cell transfusion (units)	5 ± 4	15 ± 13	0.002	
Fresh frozen plasma (units)	7 ± 5	20 ± 8	0.006	
Platelets (units)	1.6 ± 1	1.3 ± 1	0.817	
Metabolomic quantification				
Alanine	1.512 ± 0.2	2.595 ± 1.5	0.175	
Valine	0.21 ± 0.04	0.528 ± 0.4	0.111	
Isoleucine	0.145 ± 0.03	0.325 ± 0.2	0.139	
Glutamate	1.795 ± 1	2.713 ± 2	0.078	
Glutamine	0.76 ± 0.4	1.087 ± 0.7	0.115	
GABA	0.485 ± 0.1	1.059 ± 0.5	0.069	
Glucose	6.697 ± 3	7.191 ± 5	0.762	
Lactate	6.122 ± 3	13.445 ± 7	< 0.0001	
Lactate >8.4 mmol/g	7 (%)	6 (%)	0.0006	0.046
Glycerol	4.699 ± 3	6.615 ± 7	0.27	
Ascorbate	0.175 ± 0.02	0.195 ± 0.04	0.702	
Glutathione	0.41 ± 0.07	0.346 ± 0.2	0.725	
Creatine	0.5 ± 0.05	0.7 ± 0.1	0.111	
Choline	1.062 ± 0.1	1.568 ± 0.6	0.17	
Phosphocholine	0.471 + 0.2	0.741 + 0.2	0.006	
Phosphocholine $>0.65 \text{ mmol/g}$	7 (%)	6 (%)	0.00006*	0.207
Glycerophosphocholine	1.18 + 0.1	1.564 + 0.5	0.15	5.207
Taurine	3.48 + 1	5.031 ± 0.5	0.11	
Ethanolamine	0.371 +	0.633 +	0.108	
	0.06	0.19	0.100	

BMI, body mass index; GABA, gamma-aminobutyric acid; GGT, gamma-glutamyl transferase; HCV, hepatitis C virus; INR, internal normalized ratio; LT, liver transplant; MELD, model for end-stage liver disease.

According to the European Association for the Study of the Liver definition.



Fig. 4. Rate of EAD according to the metabolic profile of the graft. The population was evaluated as a whole (n = 42), or according to the presence (light blue) or not (dark blue) of sarcopenia. This graph shows the significant impact of metabolic donor-recipient matching as use of lactate and/or phosphocholine rich-liver grafts in patients with sarcopenia, who exert low lactate and glycerophosphocholine content, is associated to a growing risk of EAD. EAD, early allograft dysfunction; L, lactate; PC, phosphocholine.

Metabolomic profile from native liver do not predict EAD

Metabolomic profiles from native liver were not significantly different between patients presenting with EAD vs. no EAD ($Q^2 = 0.296$, $R^2\gamma = 0.629$).

Donor-recipient matching according to MD-ECD

Given its significant impact on liver metabolic profile, sarcopenia was used as a major determinant for donor-recipient matching. Use of a MD-ECD in a patient with sarcopenia was associated with a 100% risk of EAD, whereas use of a non MD-ECD in a patient without sarcopenia was associated with no risk of EAD. Most interestingly, EAD was more common in recipients with sarcopenia, receiving grafts from phosphocholine-poor donors underlining the metabolic cost of graft adaptation (Fig. 4).

Discussion

Evaluation of graft quality is a highly complex daily task for transplant teams, as many factors must be considered. Metabolic biomarkers have been evaluated, but they may provide the unique advantage of informing clinicians on the cellular state and function of a graft. This study demonstrates that metabolomics using HR-MAS-NMR is efficient in predicting EAD and is potentially applicable to daily clinical practice.

The predictive value of metabolomic study of liver grafts has already been shown in a previous study using mass spectroscopy.¹⁶ HR-MAS-NMR presents the advantage of being easily applicable in the clinical setting and the data from this study support its use in a clinical setting

One of the main findings of this study was to show that lactate content is highly predictive of EAD, even when measured as early as back-table preparation. Given the 0.906 AUROC, lactate content is one of the most powerful reported biomarkers for prediction of early graft outcome. Indeed, even established scores using both donor and recipient parameters do not achieve such accuracy. For instance, the BAR score achieves a c-statistic of 0.7.²⁷ Although adding the degree of macrovascular steatosis increases the accuracy, the simple measure of lactate content, available at the time of back-table preparation, is at least as powerful as the modified BAR score.

The high level of lactate is mainly due to anaerobic metabolism that takes place during cold ischemia. It may be a marker of graft tolerance to static cold storage and underlines the impact of mitochondrial dysfunction during this phase. Recent works have shown the beneficial impact of oxygenation on mitochondrial function in liver perfused grafts.²⁸ In mammalian cells, lactate represents the end product of anaerobic glycolysis, explaining its significant impact as a predictive biomarker for liver dysfunction. High lactate levels induce local acidosis and may act as damage-associated molecular patterns, leading to exacerbated inflammatory response at reperfusion.²⁹ Furthermore intracellular acidosis and cold storage may lead to imbalance in the protein turnover^{30,31} that lead to vital protein destruction and enzymatic inhibition that will block the reconstitution of the ATP reserve after reperfusion. Systemic lactate clearance within the 24 h after liver transplantation is a marker of graft function.³² The lactate content in the graft should be distinguished from excreted lactates which are measured in the effluent. Another metabolomic study on machine perfused grafts has shown that lactate effluent may be low even in cases of high intragraft content.

Glutamine is the most important amino acid in the body with major roles in gut integrity and immune system viability, as it can serve as an energy source through the production of α -ketoglutarate, one of the components of the Krebs cycle. It is also the main carrier of ammonium. Glutamate is the intermediate metabolite between glutamine and α -ketoglutarate. Elevation of glutamate has already been reported in metabolomic studies of cirrhotic livers, according to the stage of liver fibrosis, necroinflammatory activity and steatosis,³³ as well as in circulating blood of diabetic patients.³⁴ In hibernating animals, high glutamine levels are observed during the hypothermic torpor period and are thought to be a metabolic adaptation for the storage of nitrogen. Indeed, during a period of hypothermic ischemia, the urea cycle is inhibited, thus limiting the capacity to eliminate ammonia.³⁵ Regulation of the glutamine-glutamate axis is not fully described, but data have shown that it is dependent on the response to steroids.³⁶

Phospholipid metabolism is highly impaired during ischemia. The formation of "blebs" in ischemic livers is a well described phenomenon preceding membrane rupture and cell destruction due to ATP depletion.³⁷ The differential concentration of choline derivatives emphasizes the major impact of membrane turnover in liver grafts subjected to ischemia. Phosphocholine is derived from choline and is transformed in endoplasmic reticulum (ER) into glycerophosphocholine (GPC). High phosphocholine may be a marker of ER stress. GPC has already been shown to decrease during the different stages of liver transplantation from procurement to reperfusion.¹⁵ GPC has also been reported as protective against microvascular alterations in rat models of ischemia.³⁸ Choline deficiency represents a model of liver disease and specifically metabolic and steatotic disease.³⁹ In mice with choline deficiency, tolerance to hepatectomy is reduced and the inflammatory response has been reported to be increased.⁴⁰

The second highlight of this study is to show the significant cost of metabolic adaptation in high risk patients. Sarcopenia is now a recognized risk factor for poor LT outcomes. Sarcopenia is a marker of long-term evolving cirrhosis and is often associated with portal hypertension and ascites. It is a sign of late

JOURNAL OF HEPATOLOGY

metabolic adaptation in a patient with cirrhosis when the liver metabolism has switched to lactate use from the muscle, by the Cori cycle. It has been well shown that end-stage liver disease is associated with a metabolic shift, whatever the cause of cirrhosis. There are now studies showing the significant difference in metabolic profiles between fibrotic, compensated cirrhotic and decompensated cirrhotic livers. Although the difference in metabolic profiles between patients with and without sarcopenia is not surprising, the striking finding of this study was that lactate and choline derivatives are decreased in patients with sarcopenia, even in cases of low MELD.

When using the GMS, it becomes obvious that donorrecipient matching is particularly important in a cirrhotic population. Although we identify patients with sarcopenia in this study, other clinical parameters may be associated with low GPC and lactate. In this population, the use of MD-ECD leads to a significant need for metabolic adaptation for the liver graft. The cost of metabolic adaptation may explain poor early graft function that is otherwise poorly explained by the routinely used clinical and biological parameters.

The main limitation of this study may be that EAD should not necessarily lead to refusing a liver graft. Although it is associated with long-term results, the life-saving place of liver transplantation may overcome this point. Although the rate of donors exerting extended criteria as defined by EASL was quite high, the rate of severely steatotic livers was low in this series. One could question the significant impact of the two steatotic liver donors on metabolic profiles as they both experienced EAD. An additional analysis confirmed the data, even after exclusion of these two donors, even with an high AUROC for lactate of 0.936. Unfortunately, it is not possible to identify steatotic livers with favorable metabolic profiles. Indeed, the literature has shown a significantly higher rate of EAD in severely steatotic livers but "only" 40% exert EAD and recent studies have confirmed good results with steatotic livers when the recipient is well selected. Another limitation of the study is the size of the population which should lead to cautious interpretation of the data. However, the studied population was representative of the overall population transplanted during the study period (n = 134). Moreover, the statistical analysis used to identify discriminating metabolites, specifically ADEMA, is a highly powerful method enabling strong conclusions even in case of small samples.²⁵

In clinical practice, the back-table time seems the most appropriate time for the study, as it takes into consideration the impact of cold ischemia on the graft, but leaves sufficient time to change strategy in case of major metabolic abnormalities on the profile. Future studies should focus on metabolic changes over time. Altogether, the data acquired through this technique, in less than 30 min, may increase the objectivity of graft selection in liver transplantation. Besides, evaluation of DCD grafts may be an excellent application of this technique as significant differences in metabolic profiles and metabolic adaptation between DCD and DBD have already been shown.⁴¹ Identifying metabolic biomarkers in this specific population may enable use of older donors and donors with longer ischemic time. In the field of lung transplantation, HR-MAS-NMR metabolomics has shown that perfusion confers significant metabolic benefit according to different ischemia times.⁴²

We advocate the use of HR-MAS-NMR metabolomics to evaluate the quality of liver grafts after cold storage. In the case of favorable metabolic profiles, liver transplantation could be safely performed, even with ECD. In light of this analysis, interpretation of biological results may avoid the question of relisting in cases where PNF is deemed a risk. In case of significant metabolic derangements identified by HR-MA-NMR, the use of machine perfusion could be indicated as there is a significantly higher risk of one-year graft loss. Most importantly the use of lactate- and phosphocholine-rich donor grafts in recipients with sarcopenia and more generally in patients with cirrhosis, exerting advanced stage metabolic derangements, should be avoided as it leads to significant graft failure. Metabolomic profiling may help to evaluate the efficiency of graft resuscitation on machine perfusion and to objectively select DCD grafts.

Conflict of interest

The authors who have taken part in this study declared that they do not have anything to disclose regarding funding or conflict of interest with respect to this manuscript.

Please refer to the accompanying ICMJE disclosure forms for further details.

Authors' contributions

Study concept and design: IJN, FF. Acquisition of data: FF, SB, ER, PA. Analysis and interpretation of data: IJN, FF, SB, ER. Statistical analysis: FF, SB, ER. Drafting of the manuscript: FF, IJN. Critical revision of the manuscript: FF, SB, CB, ER, BE, MLWJ, PA, PB, IJN.

Supplementary data

Supplementary data associated with this article can be found, in the online version, at https://doi.org/10.1016/j.jhep.2017.11. 022.

References

- Ghinolfi D, Marti J, De Simone P, Lai Q, Pezzai D, Coletti L, et al. Use of octogenarian donors for liver transplantation: a survival analysis. Am J Transplant 2014;14:2062–2071.
- [2] Dutkowski P, Schlegel A, Slankamenac K, Oberkofler CE, Adam R, Burroughs AK, et al. The use of fatty liver grafts in modern allocation systems; risk assessment by the balance of risk (BAR) score. Ann Surg 2012;256:861–869.
- [3] Chu MJ, Dare AJ, Phillips AR, Bartlett AS. Donor hepatic steatosis and outcome after liver transplantation. J Gastrointest Surg 2015;19:1713–1724.
- [4] Burroughs AK, Sabin CA, Rolles K, Delvart V, Karam V, Buckels J, et al. 3-Month and 12-month mortality after first liver transplant in adults in Europe: predictive models for outcome. Lancet 2006;367:225–232.
- [5] Gruttadauria S, Vizzini G, Biondo D, Mandalà L, Volpes R, Palazzo U, et al. Critical use of Extended Criteria Donor Liver grafts in adult-to-adult whole liver transplantation: a single-center experience. Liver Transpl 2008;14:220–227.
- [6] Renz JF, Kin C, Kinkhabwala M, Jan D, Varadarajan R, Goldstein M, et al. Utilization of Extended Donor Criteria Allografts maximizes donor use and patient access to liver transplantation. Ann Surg 2005;242:556–563.
- [7] Ghinolfi D, De Simone P, Lai Q, Pezzati D, Coletti L, Balzano E, et al. Risk analysis of ischemic-type biliary lesions after liver transplant using octogenarian donors. Liver Transpl 2016;22:588–598.
- [8] Ali JM, Davies SE, Brais RJ, Randle LV, Klinck JR, Allison ME, et al. Analysis of ischemia/reperfusion injury in time-zero biopsies predicts allograft outcomes. Liver Transpl 2015;21:487–499.
- [9] Gaffey MJ, Boyd JC, Traweek ST, Ali JM, Rezeig M, Caldwell SH, et al. Predictive value of intraoperative biopsies and liver function tests for preservation injury in orthotopic liver transplantation. Hepatology 1997;25:184–189.
- [10] Dutkowski P, Schlegel A, Slankamenac MC, Oberkofler CE, Adam R, Burroughs AK, et al. The use of fatty liver grafts in modern allocation systems: risk assessment by the balance of risk (BAR) Score. Ann Surg 2012;256:861–869.

Research Article

- [11] Gabrielli M, Moisan F, Vidal M, Duarte I, Jiménez M, Izquierdo G, et al. Steatotic livers. Can we use them in OLTX ? Outcome data from a prospective baseline biopsy study. Ann Hepatol 2012;11:891–898.
- [12] El-Badry AM, Breitenstein S, Jochum W, Washington K, Paradis V, Rubbia-Brandt L, et al. Assessment of hepatic steatosis by expert pathologists : the end of a gold standard. Ann Surg 2009;250:691–697.
- [13] Monbaliu D, Liu Q, Libbrecht L, De Vos R, Vekemans K, Debbaut C, et al. Preserving the morphology and evaluating the quality of liver grafts by hypothermic machine perfusion: a proof-of-concept studyusing discarded human livers. Liver Transpl 2012;18:1495–1507.
- [14] Bruinsma BG, Yeh H, Ozer S, Martins PN, Farmer A, Wu W, et al. Subnormothermic machine perfusion for ex vivo preservation and recovery of the human liver for transplantation. Am J Transpl 2014;14:1400–1409.
- [15] Op den Dries S, Karimian N, Sutton ME, Westerkamp AC, Nijsten MW, Gouw AS, et al. Ex vivo normothermic machine perfusion and viability testing of discarded human donor livers. Am J Transpl 2013;13:1327–1335.
- [16] Moers C, Smits JM, Maathuis MH, Treckmann J, van Gelder F, Napieralski BP, et al. Machine perfusion or cold storage in deceased-donor kidney transplantation. N Eng J Med 2009;360:7–19.
- [17] Graham JA, Guarrera JV. Resuscitation of marginal liver allografts for transplantation with machine perfusion technology. J Hepatol 2014;61:418–431.
- [18] Okamura Y, Hata K, Tanaka H, Hirao H, Kubota T, Inamoto O, et al. Impact of subnormothermic machine perfusion preservation in severely steatotic rat livers: a detailed assessment in isolated setting. Am J Transpl 2017;17:1204–1215.
- [19] Duarte IF, Stanley EG, Holmes E, Lindon JC, Gil AM, Tanq H, et al. Metabolic assessment of Human Liver Transplants from biospy samples at the donor and recipient stages using High-resolution Magic angle Spinning ¹H NMR spectroscopy. Ann Chem 2005;77:5570–5578.
- [20] Cortes M, Pareja E, Garcia-Canaveras JC, Donato MT, Montero S, Mir J, et al. Metabolomics discloses donor liver biomarkers associated with early allograft dysfunction. J Hepatol 2014;61:564–574.
- [21] Memeo R, Le Roy B, Pittau G, Ntona E. N. de'Angelis. Personal experience with 150 isolated procurements of liver allografts: two different techniques. Prog transplant 2015;25:287–288.
- [22] European Association for the study of the liver. EASL clinical practice guidelines: liver transplantation. J Hepatol 2016;64:433–485.
- [23] Olthoff KM, Kulik L, Samstein B, Kaminski M, Abecassis M, Emond J, et al. Validation of a current definition of early allograft dysfunction in liver transplant recipients and analysis of risk factors. Liver Transpl 2010;16:943–949.
- [24] Golse N, Bucur PO, Ciacio O, Pittau G, Sa Cunha A, Adam R, et al. A new definition of sarcopenia in patients undergoing liver transplantation. Liver Transpl 2017;23:143–154.
- [25] Cicek AE, Bederman I, Henderson L, Drumm ML, Ozsoyoglu G. ADEMA: an algorithm to determine expected metabolite level alterations using mutual information. PLoS Comput Biol 2013;9:e1002859.
- [26] Selway ZZ. Metabolism at a glance. 3rd ed. Malden, MI: Blackwell Publishing; 2014.
- [27] Dutkowski P, Oberkofler CE, Slankamenac K, Puhan MA, Schadde E, Müllhaupt B, et al. Are there better guidelines for allocation in liver transplantation? A novel score targeting justice and utility in the Model for End-stage liver disease era. Ann Surg 2011;254:745–753.

- [28] van Golen RF, van Gulik TM, Heger M. Mechanistic overview of reactive species-induced degradation of the endothelial glycocalyx during hepatic ischemia/reperfusion injury. Free Radic Biol Med 2012;52:1382–1402.
- [29] Land WG, Agostinis P, Gasser S, Garg AD, Linkermann A. Transplantation and Damage-Associated Molecular Patterns (DAMPs). Am J Transplant 2016;16:3322–3337.
- [30] Jaeschke H. Molecular mechanisms of hepatic ischemia-reperfusion injury and preconditioning. Am J Physiol Gastrointest Liver Physiol 2003;284:G15–G26.
- [31] Defamie V, Cursio R, Le Brigand K, Moreilhon C, Saint-Paul MC, Laurens M, et al. Gene expression profiling of human liver transplants identifies an early transcriptional signature associated with initial poor graft function. Am J Transplant 2008;8:1221–1236.
- [32] Wu JF, Wu RY, Chen J, Ou-Yang B, Guan XD. Early lactate clearance is a reliable predictor of initial poor graft function after orthotopic liver transplantation. Hepatobiliary Pancreat Dis Int 2011;10:587–592.
- [33] Sarfaraz MO, Myers RP, Coffin CS, Gao ZH, Shaheen AA, Crotty PM, et al. A quantitative metabolomics profiling approach for the non invasive assessment of liver histology in patients with chronic hepatitis C. Clin Trans Med 2016;5:33.
- [34] Cheng S, Rhee EP, Larson MG, Lewis GD, McCabe EL, Shen D, et al. Metabolite Profiling identifies Pathways associated with metabolic risk in humans. Circulation 2012;125:2222–2231.
- [35] Lindell SL, Klahn SL, Piazza TM, Mangino MJ, Torrealba JR, Southard JH, et al. Natural resistance to liver cold ischemia-reperfusion injury associated with the hibernation phenotype. Am J Physiol Gastrointest Liver Physiol 2005;288:473–480.
- [36] Abrahám I, Juhász G, Kékesi KA, Kovács KJ. Corticosterone peak is responsible for stress induced elevation of glutamate in the hippocampus. Stress 1998;2:171–181.
- [37] Lemasters JJ, DiGuiseppi J, Nieminen AL, Herman B. Blebbing, free Ca2+ and mitochondrial membrane potential preceding cell death in hepatocytes. Nature 1987;325:78–81.
- [38] Hartmann P, Fet N, Garab D, Szabó A, Kaszaki J, Srinivasan PK, et al. Lalpha-GPC reduces the microcirculatory dysfunction and nicotinamide adenine dinucleotide phosphate-oxidase type 4 induction after partial hepatic ischemia in rats. J Surg Res 2014;189:32–40.
- [39] Lau JK, Zhang X, Yu J. Animal models of non –alcoholic fatty liver disease: current perspectives and recent advances. J Pathol 2017;241:36–44.
- [40] van der Veen JN, Lingrell S, Gao X, Quiroga AD, Takawale A, Armstrong EA, et al. Pioglitazone attenuates hepatic inflammation and fibrosis in phosphatidylethanolamine-N-mehtyl-transferase-deficient mice. Am J Physiol Gastrointest Liver Physiol 2016;310:G526–G538.
- [41] Hrydziuszko O, Perera MT, Laing R, Kirwan J, Silva MA, Richards DA, et al. Mass spectroscopy based metabolomics comparison of liver grafts from donors after circulatory death (DCD) and donors after brain death (DBD) used in human orthotopic liver transplantation. PLoS One 2016;11: e0165884.
- [42] Benhamed MA, Santelmo N, Elbayed K, Frossard N, Noll E, Canuet M, et al. The assessment of the quality of the graft in animal model for lung transplantation using the metabolomics ¹H high-resolution magic angle spinning NMR spectroscopy. Magn Res Med 2012;68:1026–1038.